Author’s response to reviews

Title: Self-collected versus clinician-collected sampling for sexually transmitted infections: A systemic review and meta-analysis protocol

Authors:

Darlene Taylor (Darlene.taylor@bccdc.ca)
Carole Lunny (carole.lunny@bccdc.ca)
Tom Wong (tom.wong@phac-aspc.gc.ca)
Mel Krajden (Mel.krajden@bccdc.ca)
Linda Hoang (Linda.hoang@bccdc.ca)
Richard Lester (Richard.lester@bccdc.ca)
Neville Li (Neville.li@bccdc.ca)
Mark Gilbert (Mark.Gilbert@bccdc.ca)

Version: 2 Date: 13 August 2013

Author’s response to reviews:

The revisions made were edits by the co-authors and corresponding author to the flow of the text.